Author’s response to reviews

Title: Microparticles shed from multidrug resistant breast cancer cells provide a parallel survival pathway through immune evasion

Authors:

Ritu Jaiswal (ritu.jaiswal@uts.edu.au)

Michael Johnson (Michael.Johnson@uts.edu.au)

Deep Pokharel (Deep.Pokharel@student.uts.edu.au)

Sabna Krishnan (Sabna.RajeevKrishnan@student.uts.edu.au)

Mary Bebawy (Mary.Bebawy@uts.edu.au)

Version: 5 Date: 20 Jan 2017

Author’s response to reviews:

1. Please provide email address details for all authors on the title page of the manuscript.
   Included on title page, lines 20-24.

2. Please include details of the sources of cell lines MCF-7 and hCMEC/D3 in the Methods.
   Included in Methods section, lines 147-150.
   Also included in Acknowledgements, lines 570-571.

3. You indicate in the Availability of data and materials section that all data is included in the article, but the data are only analysis of the raw data; no raw data are included. Please clarify.
   This has no been clarified. Lines 545-547.

Other changes include: Funding section-lines 556-559.